Title: An unusual case report of metachronous squamous cell carcinoma of pancreas and stomach in an elderly female patient

Reviewer #1: The authors had reported a rare case of metachronous squamous cell carcinoma of pancreas and stomach in an elderly female patient. From the course of the disease, the prognosis of the disease is very poor. As there is no reported case of such case, it would be of interest to the readers of WJG.

Answer: Thank you for your comment.

Reviewer #2: In this report, the author showed us a rare case of metachronous gastric SCC after pancreatic SCC treatment. Results of immunohistochemistry and HE also confirmed the degree of differentiation and specificity of the tissue. This report is innovative to some extent and has reached the requirements of the magazine.

Answer: Thank you for your comment.